

NORTHAMPTON BOROUGH COUNCIL

SCRUTINY PANEL 3 DEMENTIA FRIENDLY TOWN

21 SEPTEMBER 2017

BRIEFING NOTE: BACKGROUND DATA - RELEVANT LEGISLATION

1 INTRODUCTION

- 1.1 At its inaugural scoping meeting, Scrutiny Panel 3 (Dementia Friendly Town) agreed that it would receive details of relevant Legislation to inform its evidence base.
- 1.2 Short summaries of the key points of the Legislation, for consideration by the Scrutiny Panel at its meeting on 21 September 2017, are:
 - The Care Act 2014
 - Mental Capacity Act 2005
 - Human Rights Act 1998
 - Mental Health Act 1983
 - Deprivation of Liberty Safeguards (2009)

1.3 The Care Act 2014

1.3.1 The Social Care Institute for Excellence has produced an easy read summary of the Care Act 2014, which highlights the key aspects of the Act:

"The Care Act is a new law that means councils must have an independent advocacy service for people who find it difficult to have a say in their care and services and do not have someone to help them with this. Independent means the service is not controlled by the council and only thinks about what people they are helping want.

Independent advocacy is about giving people as much control as possible in their lives. It helps people understand information, say what they want and what they need and get the services they need.

Advocates work with people as partners and are always on their side.

Advocacy helps people in general and helps everyone to live together. It helps people to be part of things, be equal with other people and get their rights.

The new law means more people are likely to use advocacy and councils will need to plan for this. This might be hard because there will be times when

there are more people using advocacy services and times when there are not so many.

Councils already have to have some advocacy services. These services are called Independent Mental Capacity Advocacy, Independent Mental Health Advocacy and NHS Complaints Advocacy. Some councils also have other advocacy services as they think they would be useful for people. Complaint means saying that something is wrong with a service and asking for it to be sorted out. When councils are thinking about what the Care Act says advocacy services they should use what is already there and work with organisations who already do advocacy.

The councils should make sure the organisations they pay to run advocacy services under the Care Act can be flexible and can give help to all the people that need it. A lot more people are likely to use advocacy services when the Care Act takes effect in April 2015.

Councils should get as much information as possible about advocacy services so they can plan for the future.

They should use this information to help them work out who should run advocacy services, how they work out who needs advocacy, how advocacy services should work and working out where new services are needed.

Training is important for everyone who is working on advocacy services including the people in the council who choose the organisations to run the services. They need to know about everything the law says the council has to do including how people first get advocacy, how advocacy works and how people get the service

It is best for councils to do all these things using co-production with local people who use services and carers. It is also important to work with organisations that run advocacy services or might run them in the future. Co-production means working together with people who use services and carers as equal partners."

 $Source: \ http://www.scie.org.uk/care-act-2014/advocacy-services/commissioning-independent-advocacy/easy-read/$

1.3.2 The Act is available.

1.4 Mental Capacity Act (MCA) 2005

1.4.1 The Social Care Institute for Excellence has produced an easy read summary of the Mental Capacity Act 2005, which highlights the key aspects of the Act:

"Key Messages

- The Mental Capacity Act (MCA) 2005 applies to everyone involved in the care, treatment and support of people aged 16 and over living in England and Wales who are unable to make all or some decisions for themselves.
- The MCA is designed to protect and restore power to those vulnerable people who lack capacity.
- The MCA also supports those who have capacity and choose to plan for their future this is everyone in the general population who is over the age of 18.
- All professionals have a duty to comply with the Code of Practice. It also provides support and guidance for less formal carers.
- The Act's five statutory principles are the benchmark and must underpin all acts carried out and decisions taken in relation to the Act.

- Anyone caring for or supporting a person who may lack capacity could be involved in assessing capacity follow the two-stage test.
- The MCA is designed to empower those in health and social care to assess capacity themselves, rather than rely on expert testing good professional training is key
- If capacity is lacking, follow the checklist described in the Code to work out the best interests of the individual concerned
- Understanding and using the MCA supports practice for example, application of the Deprivation of Liberty Safeguards

Introduction

This Act a glance summary presents an overview of the Mental Capacity Act (MCA) 2005, which is important to health and social care practice.

The MCA has been in force since 2007 and applies to England and Wales. The primary purpose of the MCA is to promote and safeguard decision-making within a legal framework. It does this in two ways:

- by empowering people to make decisions for themselves wherever possible, and by protecting people who lack capacity by providing a flexible framework that places individuals at the heart of the decision-making process
- by allowing people to plan ahead for a time in the future when they might lack the capacity, for any number
- Reach

About two million people in England and Wales are thought to lack capacity to make decisions for themselves. They are cared for by around six million people, including a broad range of health and social care staff, plus unpaid carers. Those working in health and social care include: doctors, nurses, dentists, psychologists, occupational, speech and language therapists, social workers, residential and care home managers, care staff (including domiciliary care workers), and support workers (including people who work in supported housing).

- A lack of mental capacity could be due to:
- a stroke or brain injury
- a mental health problem
- dementia
- a learning disability
- confusion, drowsiness or unconsciousness because of an illness of the treatment for it
- substance misuse.

Five key principles

The Act is underpinned by five key principles (Section 1, MCA). It is useful to consider the principles chronologically: principles 1 to 3 will support the process before or at the point of determining whether someone lacks capacity. Once you've decided that capacity is lacking, use principles 4 and 5 to support the decision-making process.

Principle 1: A presumption of capacity

Every adult has the right to make his or her own decisions and must be assumed to have capacity to do so unless it is proved otherwise. This means that you cannot assume that someone cannot make a decision for themselves just because they have a particular medical condition or disability.

Principle 2: Individuals being supported to make their own decisions

A person must be given all practicable help before anyone treats them as not being able to make their own decisions. This means you should make every effort to encourage and support people to make the decision for themselves. If lack of capacity is established, it is still important that you involve the person as far as possible in making decisions.

Principle 3: Unwise decisions

People have the right to make decisions that others might regard as unwise or eccentric. You cannot treat someone as lacking capacity for this reason. Everyone has their own values, beliefs and preferences which may not be the same as those of other people.

Principle 4: Best interests

Anything done for or on behalf of a person who lacks mental capacity must be done in their best interests.

Principle 5: Less restrictive option

Someone making a decision or acting on behalf of a person who lacks capacity must consider whether it is possible to decide or act in a way that would interfere less with the person's rights and freedoms of action, or whether there is a need to decide or act at all. Any intervention should be weighed up in the particular circumstances of the case.

Assessment of capacity and best interests decision-making (Sections 2–4, MCA)

What is mental capacity and when might you need to assess capacity?

Having mental capacity means that a person is able to make their own decisions. You should always start from the assumption that the person has the capacity to make the decision in question (principle 1). You should also be able to show that you have made every effort to encourage and support the person to make the decision themselves (principle 2). You must also remember that if a person makes a decision which you consider eccentric or unwise, this does not necessarily mean that the person lacks the capacity to make the decision (principle 3). Under the MCA, you are required to make an assessment of capacity before carrying out any care or treatment – the more serious the decision, the more formal the assessment of capacity needs to be.

When should capacity be assessed?

You might need to assess capacity where a person is unable to make a particular decision at a particular time because their mind or brain is affected by illness of disability. Lack of capacity may not be a permanent condition. Assessments of capacity should be time- and decision-specific. You cannot decide that someone lacks capacity based upon age, appearance, condition or behaviour alone.

The test to assess capacity

Two-stage functional test of capacity

In order to decide whether an individual has the capacity to make a particular decision you must answer two questions:

Stage 1. Is there an impairment of or disturbance in the functioning of a person's mind or brain? If so,

Stage 2. Is the impairment or disturbance sufficient that the person lacks the capacity to make a particular decision?

The MCA says that a person is unable to make their own decision if they cannot do one or more of the following four things:

- understand information given to them
- retain that information long enough to be able to make the decision
- weigh up the information available to make the decision
- communicate their decision this could be by talking, using sign language or even simple muscle movements such as blinking an eye or squeezing a hand.

Every effort should be made to find ways of communicating with someone before deciding that they lack capacity to make a decision based solely on their inability to communicate. Also, you will need to involve family, friends, carers or other professionals.

The assessment must be made on the balance of probabilities – is it more likely than not that the person lacks capacity? You should be able to show in your records why you have come to your conclusion that capacity is lacking for the particular decision.

Best interests decision-making

If a person has been assessed as lacking capacity then any action taken, or any decision made for or on behalf of that person, must be made in his or her best interests (principle 4). The person who has to make the decision is known as the 'decision-maker' and normally will be the carer responsible for the day-to-day care, or a professional such as a doctor, nurse or social worker where decisions about treatment, care arrangements or accommodation need to be made.

What is 'best interests'?

The Act provides a non-exhaustive checklist of factors that decision-makers must work through in deciding what is in a person's best interests. A person can put his/her wishes and feelings into a written statement if they so wish, which the person determining capacity must consider. In addition, people involved in caring for the person lacking capacity have to be consulted concerning a person's best interests."

Source: http://www.scie.org.uk/mca/introduction/mental-capacity-act-2005-at-a-glance

1.4.2 The Act is available.

1.5 Human Rights Act 1998

1.5.1 The Human Rights Act, came into full force on 2 October 2000.

The intention of introducing the Act is –

"To help create a society in which people's rights and responsibilities are properly balanced and where an awareness of the Convention rights permeates our government and legal systems at all levels." The Human Rights Act places a duty on all courts and tribunals in the UK to interpret legislation so far as possible in a way compatible with the rights laid down in the European Convention on Human Rights (section 3(1)). Where this is not possible, the court may issue a "declaration of incompatibility". The declaration does not invalidate the legislation, but permits the amendment of the legislation by a special fast-track procedure under section 10 of the Act.

The Human Rights Act applies to all public bodies within the United Kingdom, including central government, local authorities, and bodies exercising public functions. However, it does not include Parliament when it is acting in its legislative capacities.

Key aspects of the Act:

Absolute rights are those that cannot be infringed.

Limited Rights are those that can, under explicit circumstances identified in the Convention, be limited.

Qualified Rights are rights that can be interfered with if what is done:

- has a legal basis
- is necessary in a democratic society i.e. it fulfils a pressing social need, pursues a legitimate aim and is proportionate to the aims being pursued
- is related to the aim set out in the relevant article e.g. the prevention of crime, the protection of public order, health or morals

Qualified rights are:

- the right to respect for private and family life
- rights relating to religion
- the right to freedom of expression
- the right to freedom of assembly and association
- the right to the peaceful enjoyment of property
- the right to education

The Convention allows rights to be interfered with to support a democratic society.

1.6 Mental Health Act 1983 (amended in 1987)

1.6.1 The Mental Health Act is the Act which sets out when you can be admitted, detained and treated in hospital against your wishes. It is also known as being 'sectioned'. You have certain rights under the Mental Health Act, including the right to appeal and the right to get help from an advocate.

The Mental Health Act has a number of sections that contain information on:

- Your rights when you are detained in hospital against your wishes (see our legal pages on sectioning).
- Your family's rights when you are detained (see our legal pages on nearest relative).
- Your rights when you are detained in hospital and also part of the criminal justice system (see our legal pages on mental health and the courts, mental health and the police and sectioning).
- Your rights around consent to treatment when you are detained (see our legal pages on consent to medical treatment).
- Your rights when you are being treated in the community

1.6.2 The Act is available

1.7 Deprivation of Liberty Safeguards (2009)

1.7.1 The Deprivation of Liberty Safeguards were introduced in 2009. They are part of the Mental Capacity Act 2005 (MCA). They are used to protect the rights of people who lack the ability (mental capacity) to make certain decisions for themselves. The safeguards cover patients in hospitals, and people in care homes registered under the Care Standards Act 2000, whether placed under public or private arrangements.

Key Information

Key information regarding Deprivation of Liberty Safeguards as reported by the Social Care Institute for Excellence:

• "The Deprivation of Liberty Safeguards are an amendment to the Mental Capacity Act 2005. They apply in England and Wales only.

- The Mental Capacity Act allows restraint and restrictions to be used but only if they are in a person's best interests.
- Extra safeguards are needed if the restrictions and restraint used will deprive a person of their liberty. These are called the Deprivation of Liberty Safeguards.
- The Deprivation of Liberty Safeguards can only be used if the person will be deprived of their liberty in a care home or hospital. In other settings the Court of Protection can authorise a deprivation of liberty.
- Care homes or hospitals must ask a local authority if they can deprive a person of their liberty. This is called requesting a standard authorisation.
- There are six assessments which have to take place before a standard authorisation can be given.
- If a standard authorisation is given, one key safeguard is that the person has someone appointed with legal powers to represent them. This is called the relevant person's representative and will usually be a family member or friend.
- Other safeguards include rights to challenge authorisations in the Court of Protection, and access to Independent Mental Capacity Advocates (IMCAs)."

Source: http://www.scie.org.uk/publications/ataglance/ataglance43.asp

2 RECOMMENDATION

2.1 That the information provided in this briefing note informs the evidence base of this Scrutiny Review.

Tracy Tiff, Scrutiny Officer, on behalf of Councillor Cathrine Russell, Chair, Scrutiny Panel 3 Dementia Friendly Town

Author: